Author’s response to reviews

Title: Development and testing of indicators to measure coordination of clinical information and management across levels of care

Authors:

Marta Beatriz Aller (maller@consorci.org)
Ingrid Vargas (ivargas@consorci.org)
Jordi Coderch (jcoderch@ssibe.cat)
Sebastià Calero (scalerom@gencat.cat)
Francesc Cots (FCots@parcdesalutmar.cat)
Mercè Abizanda (107@pamem.org)
Joan Farré (jfarre@fssm.cat)
Josep Llopart (jrllopart@bsa.cat)
Lluís Colomés (lcolomes@grupsagessa.com)
María Luisa Vázquez (mlvazquez@consorci.org)

Version: 3  Date: 9 July 2015

Author’s response to reviews:

REF. NO.: MS 9906478441570499

Dear Mr. Evangelista,

Following your instructions, please find below the description of the changes made in the manuscript.

Changes made

1. Please add all authors’ initials to Authors’ Contribution Section.
   All the author’s initials have been added.

2. Please delete the Funding section and add funding info to Acknowledgement section.
   The Funding section has been deleted, and the information has been added to Acknowledgement section.

Yours sincerely,

Marta Beatriz Aller Hernández
Health Policy and Health Services Research Group, Health Policy Research Unit
Consortium for Health Care and Social Services of Catalonia